**Author’s response to reviews**

**Title:** Genomic characterization of a large plasmid containing a blaNDM-1 gene carried on Salmonella enterica serovar Indiana C629 isolate from China

**Authors:**
zulqarnain baloch (znbalooch@yahoo.com)
wang wei (znbalooch@hotmail.com)
Zixin Peng (zeshanfeed@gmail.com)
Yujie hu (zulqarnain@kmust.edu.cn)
jin xu (864553361@qq.com)
Fengqin li (2479705561@qq.com)

**Version:** 2  **Date:** 29 May 2017

**Author’s response to reviews:**

Dear Editor of BMC Infectious Diseases

We greatly appreciate your efforts and consideration of our manuscript for possible publication. Please convey our heartfelt thanks to all the referees for reviewing our manuscript. There are no suggestions from the referees. However, editor himself raises several importance suggestions so i have revised this manuscript carefully according to editor suggestions.

As you said that we just have to upload our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different color. Therefore we did not add any tracked changes, comments, highlights, strikethroughs or text in different color.

I revised my manuscript according to referee recommendation. However if still there is any mistake please inform me I will make it correct. Thank you again for your attention on these revisions. I look forward your last decision.
Sincerely Yours,

Dr. Zulqarnain Baloch
Assistant Prof/ postdoc
South China Agricultural University, Guangzhou